Author’s response to reviews

Title: Occurrence of colorectal aberrant crypt foci depending on age and dietary patterns of patients

Authors:
Marek Kowalczyk (forel@neostrada.pl)
Marcin Orłowski (m.orlowski@przychodniachelm.pl)
Piotr Siermontowski (nurgok@tlen.pl)
Dariusz Mucha (nauka.autograf@gmail.com)
Krzysztof Zinkiewicz (kzinek@yahoo.com)
Waldemar Kurpiewski (wkurpiewski@interia.pl)
Ewa Zieliński (ewa.zielinski.bydgoszcz@wp.pl)
Iwona Kowalczyk (iwonakowalczyk71@wp.pl)
Agnieszka Pedrycz (apw4@wp.pl)

Version: 1 Date: 24 Sep 2017

Author’s response to reviews:

Dear Professors, Reviewers,

We are very thankful for the accurate remarks to our manuscript.

In the attachment we send our revised manuscript - we made corrections in all the points indicated:

Page 5, section Methods of line 44, has been added "Colonoscope CF-Q-165L (not HD)," Fibromyalgia Fc-240U Olympus and catheter type spray Olympus were used in the entire 131 subjects underwent a full colonoscopy."

Page 6, Section 30 Methodology section, comment has been added on Fig. 4 "In the examined group the most common finding was ACF normal and the rarest one was mixed."
In patients in whose diet red meat is prevalent or it consists mostly of red meat there are more cases of ACF. Fish the number of ACF cases was lower.

Methods in the research group where people use a high-fiber diet there is a lower case number of ACFs in the group who did not follow such diets which are especially visible when the number of ACF cases is higher than 5.

The probability of higher numbers of ACFs (5 < ACF < 10 and ACF > 10) was observed in patients with colon diverticula.

In 72 people in the examined diverticula group were found in the large intestine.

Roncucci et al. (26) states that one type of cell kinetics disorder in ACF in people suffering from CRC which is the rise of total cell proliferation in ACF.

"Vaccina et al. Using scanning electron microscopy showed that the mucosal surface among the aberrant crypts was flattened because of the loss of microvilli. (40)."


In the next attachment we send corrections to all fig. - language corrections.

In the next two attachments, we send a reply to the remarks of Professor Ron Roncucci and the replies to comments by Professor Jairam Vanamal.